[image: image1.png]10
11
12
13

Bench
TBAI
TBAZ
TBE1
TEE1
TBE2
TEC1
TaC2
TED1
TBE1
TBE1
TBE2
TEF1
TEF2

Magnet

MEAR1269
MBAR2224

MBAL2133
MBER1263
555083

MBAL3413
MBAR1267
MBBL2159
MBAR2247
MBER3416

7
b

X| X| X| X| X| X| X| X[IX]| X| X| X| X

Magnets under Test from 27-11-2005 to 03-12-2005

24112005
23112006
24112005
29112005
26-11-2005

26112005

T 23112005

25112005
24112005
26112006
26-11-2005
24112005

26112005

27h

A0h
20h
3th
£

=t

23h
h
41h
£
24h
2h
24h
14h,
24h
12h.
18 h.
47h
23h

ime

16 h.
18 h.
15 h.
gh
10h.
15 h.
13h.
9h
14h,
15 h.
14h,
14h,
14h,

ime.

10h.

16 h.
16 h.
7h
29h
17h.
14h,
19h.
13h.
7h

End Date
29112005
26112005
29112005
03-12:2005
29112005
01-12:2006
27112008
01122005
26112005
0212:2006
30112008
29112005

03-12:2005

Total
118 h.
116 h.
119 h.
@h
8h
101 h.
90k
129 h,
104 .
8h
106 h.
129 h,
102 h.

Signed for
Stripping
Stripping
Stiipping
Stiipping
Sttipping
Stiipping
Stiipping
Stiipping
Stripping
Stripping
Stiipping
Stand By

Stripping



[image: image2.png]10
11
12
13

Bench
TBAI
TBAZ
TBE1
TEE1
TBE2
TEC1
TaC2
TED1
TBE1
TBE1
TBE2
TEF1
TEF2

Magnet

MEAR1269
MBAR2224

MBAL2133
MBER1263
555083

MBAL3413
MBAR1267
MBBL2159
MBAR2247
MBER3416

7
b

X| X| X| X| X| X| X| X[IX]| X| X| X| X

Magnets under Test from 27-11-2005 to 03-12-2005

24112005
23112006
24112005
29112005
26-11-2005

26112005

T 23112005

25112005
24112005
26112006
26-11-2005
24112005

26112005

27h

A0h
20h
3th
£

=t

23h
h
41h
£
24h
2h
24h
14h,
24h
12h.
18 h.
47h
23h

ime

16 h.
18 h.
15 h.
gh
10h.
15 h.
13h.
9h
14h,
15 h.
14h,
14h,
14h,

ime.

10h.

16 h.
16 h.
7h
29h
17h.
14h,
19h.
13h.
7h

End Date
29112005
26112005
29112005
03-12:2005
29112005
01-12:2006
27112008
01122005
26112005
0212:2006
30112008
29112005

03-12:2005

Total
118 h.
116 h.
119 h.
@h
8h
101 h.
90k
129 h,
104 .
8h
106 h.
129 h,
102 h.

Signed for
Stripping
Stripping
Stiipping
Stiipping
Sttipping
Stiipping
Stiipping
Stiipping
Stripping
Stripping
Stiipping
Stand By

Stripping





[image: image3.png]fa .

TR

A 17mA
oY L

e
o

B9

12024 12851 A

14 nesRzsEs

@

2 e o WA
e [ETA 0A
] (©H) (©HLS) (02.U_S01)
= A A
s ©0 UKE
I 19225 A (1247224 (128052 A
el o) o e
WeEr S |
mia meA | g
(639 (©ET
man |
UIATE
wn |
WY
1295A 1BUA °
e e
; = PNT=ONCEDN
MBI o1y (P21 |1 @
TN
o e @
DA A 0 |
MR oy " oo NIWAE
A oA I





	 hours
	Connect
	Cooling
	Cold Test
	Warming
	Disconnect

	 
	Time
	Time
	Time
	Time
	Time

	 
	 
	 
	 
	 
	 

	AVERAGE
	29
	22
	26
	13
	15

	sigma
	8
	4
	8
	2
	5
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                                                                     .E.Veyrunes

Week 49 starting 03 Dec 05
SM18 Operation Meeting (Tuesdays)

Bench status  on Monday  05 December 05 
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Coordinator this week: M.Bajko; F.Chiusano support 16-3602
13 complete Magnet Tests cycles

9 dipoles (no MM )+ 4 SSS (without MM & SSW )
   Disconnected or ROBTC after tests last week ( Sat Midnight to Sat midnight) 
· Situation from Cryogenics:  J-P Lamboy or ACR Rep.
· Situation from Mechanics:  J.Halik (SSS, SSSS).
General Points :

· CFB on TBA1: HV problem (Richard) 
· MB_1286 (Short_circuit inter-spire): investigation results 

· Special SSS605 (status, comments about MM)

· MB2202_special tests, MB3370_HP tests, 2259_HP tests.
	Date
	Bench
	Problem
	Type
	Application/

soft
	Action/advise

	Tuesday_29_N
	TBD1


	SSS083: Problem with CQR in CQW2 for MQT corrector voltage. The voltage is correct (1.56 V) but not recognized by the program which put it "BAD". 
This gives a N.C in CQW2.
	MTM
	
	OP team will correct the template with A.Stafiniak..

	Wednesday_30_M
	TBC2


	MBBR2239 AQA alerts are present: Voltage, Noise and Jump exceeded the limits 
	MTM
	AQA
	Equipment Support and Test coordinator has been called. According Vittorio and Pierre this magnet has short circuit. The magnet has been put in Stand By 

	Wednesday_30_M
	TBA1
	Problem faced in CDW1 HV test. Rack used MY93. Current value exceeded maximum specified. Run recovery tried without success. Rack changed to MY92 and test was rerun. Problem persisted. 
	MTM
	
	Richard was contacted. After investigation, I have asked ICS to disconnect this magnet from the bench with high priority. We will see if the problem remains.
HV problem is originating from the CFB side. Investigation is on progress by Richard

Back on operation Thursday and connect MB1266.

Maintenance of the switch imperative during the shut-down

	Wednesday_30_A
	TBE1
	HV test (CDAP) was generated but no file is there
	IT
	HV
	Richard is informed.

SM18 server was moved. Richard changed the server hosting the HV files: OK.

	Wednesday_30_N
	TBA2
	Tcoils values not correct coming from shaft 28 and 27.
	MTM
	
	Calibration parameters stored in MTF must be checked and also the raw voltage of T_coil


	Thursday​_01_M
	TBB1
	A random trigger is present during IAP @ cold A/C compensation.
	MTM
	
	Connector P11 not properly plugged. The connector on the bench is not conform to fix it on the bench. Giorgio asked a change in the fixation.



	Thursday​_01_M
	TBB1
	Butterfly screws to connect the orbit dipole correctors are missing.


	MTM
	
	Giorgio furnished the screws.


	Thursday​_01_A
	TBA2
	Wrong values of multipoles for sector 0607 ap1 found during the load-line on MB3370.
	MTM
	
	Stephane asked Richard to check if it is coming from the coil c of this sector.

Check of the TRU by Maryline.

	Thursday​_01_A
	TBA2
	3rd d training quench decided by Stephane for MB3370 because first quench in the straight part and no training effect for the second quench.

Test master was not working so quench to be launched manually. Localisation lost because switch box not put in Qloc.

	IT
	TEMA
	Stephane decided to remove the shafts for special tests with HP.

	Friday_02_M
	TBF2
	Cool down of the MB2202 blocked during the night because of SSW installation.

	MTM
	
	SSW can be installed during cool down.

	Friday_02_M
	TBA2
	SPA during special LHC cycle. Cryo OK was stable.

	AB/PO?
	
	

	Friday_02_A
	TBE1
	MB1287: TT821 not working.

	ACR?
	
	NC opened in CDW1.

	Friday_02_N
	TBE1
	CERNHOMES.cern.ch is offline. Hence, unable to access G:\users\s\sm18. Thus, unable to generate HV report on bench TBB2 and TBE1. 
	IT?
	
	IT support and equip.support  contacted. Solved after 1 hour.

See comments of A.Raimondo.

	Friday_02_A
	TBE1
	MBBL2239: Launched provoked quench @3kA. But unable to do AQA analysis (AQA MIITS and AQA quench detection). 
	AB/CO
	AQA
	Analysis performed in manual with hfoff.

No lf data recorded so no analyses from AQA.

	Friday_02_A
	TBD1
	Axis measurements on SSS347. XY measurement gives bad results for Y axis (X axis OK) Tried 3 times with widely different results.
	MTM
	
	Measurements at warm no requested. 

No time to investigate before the

 cool down. 

	Saturday_03_A
	TBD1
	SW13 repeated on SSS347. During tension re-adjustment the tension adjustment knob was rotating back. The wire  was broken during tension adjustment.
	MTM
	
	Marco contacted. He advised to go ahead. Guy will look after this system. 

	Saturday_03_A
	TBC2
	MB2239: Richard advised to conduct quench at 9kA.

	MTM
	
	Special investigation with provoked quench..



	Saturday_03_N
	TBA1
	MB1266 : HV dipole vs ground fault during CDBP (HV@ cold before power tests) suspected.


	MTM
	
	Richard will investigate. Pierre advised to proceed with the tests but to check again after quench at 1.5 kA.

Before CDAP, we have to disconnect the power cables (water cooled) before launching the final HV test to ensure that the magnet is good.

	Saturday_03_N
	TBF1
	SSW measurements (field direction) not working on MB2202.

	MTM
	
	Marco came Sunday and solved the problem. Measurements in Aperture 1 was correct.

	Saturday_03_N
	TBD2
	Full scan on aperture 1: SPA because of loss of cryo OK. Measurements at 1.250 mbar (minimal pressure)

	ACR
	
	Stephane advised to repeat all the cycle at 1350 mbar.


Training performance.





Long training





D1








